PEER-REVIEW REPORT

Name of journal: World Journal of Clinical Cases
Manuscript NO: 91877
Title: Therapeutic effect of Wendan Decoction combined with mosapride on gastroesophageal reflux disease after esophageal cancer surgery
Provenance and peer review: Unsolicited Manuscript; Externally peer reviewed
Peer-review model: Single blind
Reviewer’s code: 07916480
Position: Peer Reviewer
Academic degree: MD
Professional title: Research Assistant
Reviewer’s Country/Territory: Italy
Author’s Country/Territory: China
Manuscript submission date: 2024-01-25
Reviewer chosen by: AI Technique
Reviewer accepted review: 2024-01-29 07:14
Reviewer performed review: 2024-02-05 09:08
Review time: 7 Days and 1 Hour

<table>
<thead>
<tr>
<th>Scientific quality</th>
<th>[ ] Grade A: Excellent</th>
<th>[ ] Grade B: Very good</th>
<th>[Y] Grade C: Good</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>[ ] Grade D: Fair</td>
<td>[ ] Grade E: Do not publish</td>
<td></td>
</tr>
</tbody>
</table>

<table>
<thead>
<tr>
<th>Novelty of this manuscript</th>
<th>[ ] Grade A: Excellent</th>
<th>[Y] Grade B: Good</th>
<th>[ ] Grade C: Fair</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>[ ] Grade D: No novelty</td>
<td></td>
<td></td>
</tr>
</tbody>
</table>

<table>
<thead>
<tr>
<th>Creativity or innovation of this manuscript</th>
<th>[ ] Grade A: Excellent</th>
<th>[Y] Grade B: Good</th>
<th>[ ] Grade C: Fair</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>[ ] Grade D: No creativity or innovation</td>
<td></td>
<td></td>
</tr>
</tbody>
</table>
**SPECIFIC COMMENTS TO AUTHORS**

In this study, the authors evaluated the therapeutic effects of Wendan Decoction, a Chinese medicine, combined with mosapride on gastroesophageal reflux disease after esophageal cancer surgery. The study is interesting, and well performed. The methods are described in detail. The results are well discussed with updated references. Overall, the manuscript is well written, and I recommend to accept this manuscript only after a minor editing. Thank you.

Reviewer #1:

**Specific Comments to Authors:**

In this study, the authors evaluated the therapeutic effects of Wendan Decoction, a Chinese medicine, combined with mosapride on gastroesophageal reflux disease after esophageal cancer surgery. The study is interesting, and well performed. The methods are described in detail. The results are well discussed with updated references. Overall, the manuscript is well written, and I recommend to accept this manuscript only after a minor editing. Thank you.
We have carefully checked our manuscript and corrected the typos and grammar errors we found. Thank you for your time and effort.